\begin{table}%t1
%\centering
\par
\caption{\label{astrometry}Published coordinates of He~3--759, including
astrometry from Tycho-2.}
\small
\begin{tabular}{lll}
\hline\hline
$\alpha$ (J2000) & $\delta$ (J2000) & Reference\\
\hline
12 11.3 & --62 29 & \citet{r62}; \citet{h76} \\
12 11.3 & --62 30 & \citet{ch79} \\
12 12 08.56 & --62 29 00.6 & \citet{t94}\\
12 12 08.6 & --62 29 01 & \citet{d01} \\
12 11 18.54 & --62 29 43.5 & Tycho-2 \\
\hline
\end{tabular}
\end{table}